Author's response to reviews

Title: Multiple Intracranial Arterial Stenosis Influences the Long-term Prognosis of Symptomatic Middle Cerebral Artery Occlusion

Authors:

Lili tian (ltian28@gmail.com)
Xuanye Yue (yuexy2002@hotmail.com)
Gangming Xi (xgmsys@163.com)
Youmeng Wang (fyswym@163.com)
Zongyou Li (fyszy1@163.com)
Ying Zhou (fysyyzy@163.com)
Gelin Xu (xfliu2@vip.163.com)
Xinfeng Liu (xfliu2@vip.163.com)

Version: 5 Date: 16 April 2015

Author's response to reviews: see over
Dear editor,

Thank you for processing our manuscript timely. According to the comments, we have revised the manuscript and submitted it again. A list of the changes made according to reviewers’ comments is attached.

Thank you very much for your help.

Lili Tian
Editor's comment:

We are pleased to inform you that the editors have now decided to accept your manuscript. However, before we can proceed we will need you to make some changes to your manuscript. We strongly urge you to make these changes promptly, as we cannot proceed to the next process until we have received a version containing the changes.

- Requesting name of ethics committee:

Please update your ethics statement to include the name of the ethics committee that approved your study.

Responding: The ethics statement have been updated in the revised edition, which included the name of ethics committee that approved this study.